

A Service of

ZBШ

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Angelici, Marta; Profeta, Paola

Working Paper Smart-Working: Work Flexibility without Constraints

CESifo Working Paper, No. 8165

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Angelici, Marta; Profeta, Paola (2020) : Smart-Working: Work Flexibility without Constraints, CESifo Working Paper, No. 8165, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: http://hdl.handle.net/10419/216561

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Smart-Working: Work Flexibility without Constraints

Marta Angelici, Paola Profeta



Impressum:

CESifo Working Papers ISSN 2364-1428 (electronic version) Publisher and distributor: Munich Society for the Promotion of Economic Research - CESifo GmbH The international platform of Ludwigs-Maximilians University's Center for Economic Studies and the ifo Institute Poschingerstr. 5, 81679 Munich, Germany Telephone +49 (0)89 2180-2740, Telefax +49 (0)89 2180-17845, email office@cesifo.de Editor: Clemens Fuest https://www.cesifo.org/en/wp An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com

- from the RePEc website: <u>www.RePEc.org</u>
- from the CESifo website: <u>https://www.cesifo.org/en/wp</u>

Smart-Working: Work Flexibility without Constraints

Abstract

Does removing the constraints of time and place of work increase the utility of workers and firms? We design a randomized experiment on a sample of workers in a large Italian company: workers are randomly divided into a treated group that engages in flexible space and time job (which we call "smart-working") one day per week for 9 months and a control group that continues to work traditionally. By comparing the treated and control workers, we find causal evidence that the flexibility of smart-working increases the productivity of workers and improves their well-being and work-life balance. We also observe that the effects are stronger for women and that there are no significant spillover effects within workers of a team.

JEL-Codes: J160, J220, J240, L200, M540.

Keywords: randomized control trial, productivity, work-life balance, well-being.

Marta Angelici Bicocca University & Dondena Italy – 20126 Milan marta.angelici@unimib.it Paola Profeta* Bocconi University & Dondena Italy – 20136 Milan paola.profeta@unibocconi.it

*corresponding author

March 2020

We thank Vittoria Dicandia, Annarita Macchioni and Giovanna Mazzeo Ortolani for research assistance. We are grateful to Massimo Anelli, Nicoletta Balbo, Francesco Billari, Marco Bonetti, Marina Brollo, Heejung Chung, Marilisa D'Amico, Daniela Del Boca, Giovanni Fattore, Vincenzo Galasso, Nicola Gennaioli, Anne-Marie Jeannet, Marco Manacorda, Monica Parrella and Joanna Rickne for their suggestions. We also thank participants of ELENA, AisP, Alp-Pop, EGEN-Stockholm conferences, the worksop on gender and economics at the University of Luxembourg, seminars at Bocconi University, Dondena, Discont and the University of Bologna. Part of the data was collected under the ELENA project of Italy's Department for Equal Opportunities and the Dondena Center of Bocconi University, financed by the Rights, Equality and Citizenship (REC) Programme of the European Union-JUST/2014/RGEN/AG/GEND/7803. This study is registered in the AEA RCT Registry and the unique identifying number is AEARCTR-0002979. All errors are ours.

1 Introduction

The outbreak of the 2019 novel coronavirus is threatening the growth of several parts of the world, including China, the world's fastest growing major economy, and Lombardy, the most productive Italian region. To contain the spread of the coronavirus and curb the contagion, a new organizational model of work, called "smart-working", is becoming increasingly important: workers can work outside their workplace and with a flexible time schedule, thanks to the use of the technology. The flexibility of where and when to work is used today to continue the work activities and avoid the collapse of the economy.

Smart-working is a fully flexible work arrangement, with the capacity to adapt quickly and intelligently to different situations. Smart workers agree with their supervisors to perform their work activities for a defined period of time outside of the company's physical workplace and according to a personalized time schedule. During this period, there are no specific constraints on the time or location of work. Due to the use of technology, smart workers may perform the same duties and activities as those of ordinary workers and achieve the same set targets and results while choosing a workplace and time schedule that are more convenient for both the activity to be performed and their personal needs. Time and space flexibility creates a new work organization, which is based upon results rather than workplace presence and work during particular hours.

Despite its massive use in recent weeks, we still know very little about the economic effects of smart-working in normal times. The available evidence, which consists only of case studies, management surveys on specific samples of workers and ex post descriptive analyses, does not allow us to appropriately identify the economic effects of smart-working. This paper fills this research gap and provides causal evidence that smart working is economically desirable. We design a randomized experiment to show that removing the constraints on

the time and space of work without altering the wages, for a limited period of working activity, improves the productivity, well-being and work-life balance of individuals.¹

Work flexibility is not a new concept. The majority of European employees (3 out of 4 on average) have access to some work flexibility. According to a recent survey conducted on US professionals (Dean and Auerbach, 2018), 47% of the interviewees reported having work flexibility. However, almost all the workers in the sample (96%) said that they want flexibility. A recent report by Gallup (Gallup, 2017) based on interviews with more than 195,600 employees found that flexibility plays a major role in an employee's decision to take or leave a job.

Work flexibility is a multidimensional concept. The traditional practice of working from home under the same wage conditions and control of the employer, which is known as telecommuting, is now used by approximately 17% of workers in Europe (Eurofound, 2017) and 16% of workers in the US (US Bureau of Labor Statistics, 2019). In their experiment on Chinese employees in a call-center (a routine job), Bloom et al. (2014) show that telecommuting can be beneficial for employees' productivity and their work-life balance, although at the cost of feeling isolated, which, in the long run, may risk reducing the benefits of this practice. However, telecommuting is only one way and one dimension of work flexibility, which is mainly based on replacing the workplace with the home, but maintaining the rigid control of the employer on the location of the work and the precise hours. Telecommuting is compatible with a limited number of jobs, mainly routine jobs. New and more complex forms of flexibility have begun to spread, including flexible location and flexible work times. In the US, approximately 52% of office employees have some choice over their work times (Gallup, 2017). Approximately one-third of the employees in Europe can choose between several fixed working schedules or independently set their working hours. These individuals are more likely to be employees with university degrees who are working in high-skilled jobs.

¹In our experiment, we introduce one day per week of smart-working, which is the standard use in normal times. The current health emergency requires smart-working for entire weeks, making our results not directly comparable with the current situation.

Both flexible place and flexible time are highly appreciated by workers. According to the Sixth European Survey on Working Conditions carried out by the European Foundation for the Improvement of Living and Working Conditions (Eurofound, 2017), workers appreciate having control and freedom over where and when to work, i.e., the possibility of working without the control of the employer at a place that is different from both the office and home, as well as a flexible time schedule. More than 20% of the workers (men and women) interviewed by the survey reported that their working hours do not fit with their family and social commitments. While working from home does not change this perception in a statistically significant way, having some freedom to set one's start and finish times and arrange breaks during the working day increases the perception that one's working hours fit in with their family and social commitments by approximately 20%. The report by Gallup confirms that the flexibility of hours is of growing importance and suggests that one's home is only one possible alternative to the workplace. Approximately 37% of the surveyed employees declared they would change their job for benefits related to a flexible working location (for part of their working week), and more than half of office workers (54%) said that they would leave their job for one that offers flexible work time (Gallup, 2017). Among the millennials, these reported percentages increased to 50% and 63%, respectively. Millennials want benefits that are directly related to their lives and those of their family members, and they are willing to switch jobs to secure them. Flexible work locations and hours are for them a priority, which is a fact that will impose a re-organization of work for employers wanting to compete for a modern workforce.

After decades of experience with telecommuting, there is a growing consensus among major international organizations, such as the OECD and the European Commission (OECD, 2016), that an effective improvement of work-life balance and productivity, which are the two major goals of work organizations, passes through more complex and different flexible work arrangements, based on the removal of the constraints on the location and scheduling of work.

Flexible work arrangements are introduced through individual or collective

bargaining, which, in several countries, are regulated by laws (Hegewisch et al., 2009). Such laws give the employees the "right to request" flexible working. In some cases, this is targeted specifically at the parents of young children (Australia, Finland, Norway and Sweden); in others, it is guaranteed to all employees, irrespective of the reason (Belgium, France, Germany, Netherlands, and the United Kingdom). Employees can also appeal to the courts in case employers refuse such a request. The "right to request" legislation in New Zealand, the United Kingdom and the Netherlands covers flexible working rights for employees in a comprehensive manner, including the scheduling of hours and the location of work. Recently, in 2017, Law 81 in Italy introduced an appropriate regulatory framework for the implementation of smart-working, defined as a "new method of forming a subordinated employment relationship without precise constraints on time or location of work and with the use of technological tools in the workers' duties and activities". Italy provides an interesting context for our analysis; while the country is characterized by a general low flexibility in work organization, firms started to show some level of interest in smart-working as early as ten years ago, well before Law 81, although this approach was limited to very small groups of workers (typically, fewer than ten).

Smart-working is associated with a trade-off. On the one hand, there are potential gains from the flexible work locations and hours, which go beyond those associated with telecommuting. By working from home, telecommuting allows workers to reduce their commuting costs and firms to optimize their costs. The reduction in costs is higher with smart-working than telecommuting, since the last one requires inspections and a constant monitoring of the workers at distance.² Moreover, home is only one possible alternative to the office, and not necessarily the more convenient alternative; the conflict between work and family may even become more visible when employees work

 $^{^{2}}$ Firms may reduce lighting costs, summer and winter climatization costs, corporate canteen costs, cleaning costs, etc. In some cases, the place of work itself becomes "smart" as offices become flexible spaces where workers perform part of their activities and have free access to all technologies; such spaces often have novel physical layouts, including, e.g., mindfulness zones and areas for team-working and communicating. The extreme, though not common, case is the "no fixed desk" office.

from home for caring purposes. It may also be the case that, for the same reduction of commuting time, the double gain of improving work-life balance and increasing productivity may be better obtained when workers work at a library, at a park, at a difference place close to their residence, or at a location that may change without the control of the employer, instead of home, where their family duties may interfere with their job activity. Moreover, removing the fixed daily start and finish times gives employees the possibility of better managing their time according to their preferences; they can enjoy long or short breaks for personal or family reasons, and they can adapt their work hours to life changes without altering their compensation. This increases their satisfaction and work-life balance, which ultimately makes this arrangement desirable to workers. In parallel, firms may optimize by rewarding these employees based on effective productivity rather than on the particular hours worked. Firms may also gain from the retention of talent and the reduction of days of absence, thus increasing their competitiveness. Additionally, time flexibility in the labor market for all workers (men and women) contributes to reducing the rewards of long hours, work at particular hours and inflexible schedules, which are considered a major driver of gender pay gaps (Bertrand, 2018) and may thus represent a step towards the "last chapter of the grand gender convergence" (Goldin, 2014).

On the other hand, smart working raises concerns about the organizational process, the productivity of workers and their well-being. Some of these concerns are shared with the telecommuting experience; for example, working outside the workplace may reduce the commitment of workers, who can then take advantage of the flexibility to take part in activities different from work. Moreover, by reducing interactions between workers and between workers and supervisors, there is a risk of a reduction in productivity, particularly in jobs with high interactions. Finally, blurring the boundaries between work and home may increase the hours of overtime, the levels of employee stress and worsen work-life balance. These concerns are even stronger in the case of smart working, when the location of work can be changed by the employee without the control of the employer. Moreover, the lack of rigid daily start and finish times can amplify the reduction of worker commitment, reduce their regular activity (in absence of strict rules on time) and increase the risk of overworking.

How the introduction of smart work addresses this trade-off is an open question that we address empirically. We design a randomized experiment to study the causal effects of the introduction of smart working in a large traditional company in the multi-utility sector in Italy. The company has never used this approach before. Following the methodology of randomized control trials (RCTs), we select a sample of 310 workers (containing both white- and blue-collar workers) and randomly divide it into two groups; the workers in the first group (the treated group) have the option to work "smart" (i.e., with no constraints on the place or time) one day per week for 9 months in agreement with their supervisors, while the workers in the second group (the control group) continue to work traditionally. We are interested in three major outcomes: productivity, well-being and work-life balance. We use objective measures of workers' performance calculated monthly by the firm (e.g., the number of dossiers processed during the month) and the number of days of leave of each worker. We complement this information with questionnaires administered to each worker and to his/her supervisor both before and after the treatment. The questions posed in the questionnaires capture several dimensions of self-assessed productivity, well-being and work-life balance. Given the randomization of the two groups, we are able to identify the causal effect of the treatment on our outcomes of interest.

Our results show that, for the same number of hours of work, workers involved in smart working increase their productivity compared to that of workers who continue working traditionally; this outcome is true whether productivity is captured by an objective measure or if it is measured according to several specific productivity traits (e.g., compliance with deadlines) reported by the same worker or by the supervisor. Smart workers are also more satisfied with their social life, free time and life in general. They claim to be more able to focus, make decisions, appreciate their daily activities, overcome problems and experience reduced stress and loss of sleep. Interestingly, both men and women spend more time engaged in household and care activities. We also observe that the effects are stronger for women and that there are no significant spillover effects within workers of a team.

Our results suggest that promoting smart working is an effective way to increase productivity and improve well-being and work-life balance. Moreover, we provide evidence that by removing the rigidity related to particular hours of work, smart working may contribute to the reduction of gender gaps in the labor market (Goldin, 2014). The high flexibility of smart-working—flexible time schedules, flexible places of work and flexible periods of flexible work to be used during the workweek—makes it a very appealing option for both employers and employees of a large category of jobs.

Our results are consistent with the idea of smart-working representing the removal of a constraint that is desirable for employees and useful for employers. In the presence of rigid work hours, when workers choose their amount of work hours, they face an implicit constraint on the hours of the day that can be dedicated to the work activity. This constraint may be binding for those workers who gain utility from taking a break to adapt to their personal and family needs; such workers have to choose whether to work part-time and hence decrease their wages, to not meet their needs or to be absent from work. Any option is costly. The removal of the time constraint increases the utility of these workers such that they can still work full-time and accommodate their needs by choosing a personalized time schedule. Their increased satisfaction and better time management may also increase their productivity during working hours. To the extent that employers value the output of the workers rather than the work at specific hours, smart-working represents a net gain.

The paper is organized as follows. The next section presents a literature review, section three describes the experiment, section four presents the data, section five describes the empirical strategy, section six shows the results, section seven discusses heterogeneous and spillover effects within teams, section eight presents additional analyses and considers robustness, section nine contains the discussion and conclusion. A set of appendixes complements the analysis of the main text.

2 Literature Review

Previous studied have analyzed the relationship between different management practices and productivity (see literature reviews in Walker (1887), Leibenstein (1966), Syverson (2011), Gibbons and Henderson (2013), and Bloom and Sousa-Poza (2013)). However, few studies have performed randomized experiments that can identify the causal effects of managerial procedures on productivity. Dutcher (2012) performs lab-based experiments exploring routine and non-routine tasks with and without remote monitoring, and observes that the more routine ones are negatively affected by mimicking a home-based environment. The author conjectures that the effect depends on the lack of peer and manager effects, which have been shown to be important in lowlevel tasks in field environments by Falk and Ichino (2006), Bandiera et al. (2005), and Mas and Moretti (2009). Kelly et al. (2014) examines the impact of a work-life balance training program randomized across branches of a large firm, observing significant reductions in employees' work-family conflicts, and improved family time and schedule control.

Bloom et al. (2014) perform a randomized experiment on a sample of call center employees of a large Chinese travel agency, randomly assigned to two groups: telecommuters and office workers. The researchers observe that telecommuters have higher productivity than do the other workers but, consistent with the prediction of previous studies of routine tasks, feel isolated. Thus, when telecommuters are given the opportunity to again choose between telecommuting or not, they prefer to come back to their workplace. This result is consistent with the evidence provided by Mas and Pallais (2017) who observe that the majority of workers of a call center do not value scheduling flexibility. Many of them, especially women with young children, instead value working from home. These papers consider a very specific work environment, namely, call centers, where all workers perform similar and routine-based tasks. However, as suggested by Dutcher (2012), work flexibility may affect the performance of routine and non-routine tasks differently. Non-routine jobs may take full advantage of flexibility, as they require a higher individual concentration than do routine tasks and are less exposed to isolation risks. However, there is no previous causal evidence of this for non-routine jobs. Moreover, as we have already emphasized, current flexible work arrangements go beyond telecommuting, by including flexible time schedule, flexible place of work and a flexible period of flexible work to be used during the workweek. Yet, there exists no causal evidence on smart-workers. Similarly, randomized experiments in firms involving a variety of job types in a developed economy are very rare.

The economic consequences of smart-working on workers' productivity have not been analyzed before, probably because productivity is difficult to measure and smart-working is a relatively new approach. Attempts to measure productivity based on objective indicators such as absenteeism (Koopman et al., 2002) and output per hour (Golden, 2012) show a positive relationship with flexible work arrangements. Self-declared productivity is also positively related to flexibility (see Riedmann et al. (2006)). However these studies provide descriptive evidence on the implementation of smart-working and may suffer from endogeneity concerns. The main reason is that they cannot control for other variables affecting productivity and cannot establish whether smartworking increases productivity, or whether companies with high productivity are more likely to introduce smart-working. The latter may also affect productivity through changes in workers' well-being and work-life balance, which are therefore important to analyze in parallel.

Labor sociologists (Kelly and Moen, 2007, Schieman et al., 2009) have studied the relationship between flexibility and work-life balance. As reviewed by Chung and Meuleman (2017), the evidence on this relationship is mixed: on the one hand, flexibility may reduce work-family conflicts (Chung, 2011, Kelly et al., 2011), while, on the other hand, it can create spillovers from work to home, blurring the boundaries between the two and increasing overtime hours of workers with negative impact on work-life balance (Golden and Wiens-Tuers, 2006). The latter effect tends to be dominant for high-skill workers in large companies, which also offer performance-related pay and other arrangements that motivate workers to work longer and harder. A gender divide emerges: while flexibility is used by women for family-friendly purposes, it is used by men for performance purposes. The evidence shows that women are more likely to stay employed after the birth of their first child and increase their capacity to work when family duties multiply and thus enjoy better work-life balance. Men instead increase their work intensity and performance-related payments with no changes in family arrangements, and earn incremental income. Thus, traditional gender roles risk being strengthened further by work flexibility. Chung (Chung, 2011) analyzes data for 28 European countries and shows that flexibility can have different impacts in different contexts: it is more beneficial for workers in job cultures with more hours worked (overtime), where men and women use it for family-friendly purposes, and the flexibility stigma is not strong. Other related outcomes, such as health (Kelly and Moen, 2007, Moen et al., 2013) and stress (Halbesleben and Buckley, 2004, Moen et al., 2016) outcomes, have been investigated, with results showing a positive relationship between schedule control and organization of work.³ There are two main drawbacks in these studies: it is difficult to claim causality, and there is no precise measurement of productivity.

The managerial implications of flexible work have also been explored in relation to performance (Leslie et al., 2012): flexibility may increase performance if workers are "happy" to control their own work schedules and work more effectively, with fewer days of sickness and leave. However, existing contributions mainly entail descriptive evidence and case studies of major companies across the world that allow smart-working, and do not provide causal evidence.

3 Experiment

We design and implement a randomized experiment to explore the effects of smart-working on productivity, well-being and work-life balance of workers. We focus on Italy, where smart-working is regulated by Law 81/2017 that includes specific provisions to encourage the use of smart-work as a way to promote work-life balance and to enhance competitiveness. The law includes protection of health and safety of workers and guarantees equal remuneration

³A large body of literature has studied nonstandard work schedules and their impact on well-being and family conflicts (Liu et al., 2011). Smart-working one day per week, however, is difficult to compare to a nonstandard schedule.

of workers. The organizational details are left to an agreement between the employer and the employee.⁴ According to the law, smart-work may be engaged in over continuous periods, on some days of the week or during some hours of the day, as agreed by the workers and the company. Personnel protection for both private- and public-sector employees is regulated by the National Institute for Insurance against Accidents at Work (INAIL).

We approached a large Italian company in the multi-utility sector and signed an agreement to design and implement smart-work as a pilot experiment. The company is listed on the Italian Stock Exchange and has 4131 workers engaged in various tasks. Workers are divided into blue- and white-collar types. Workers perform non-routine tasks. Blue-collar workers perform tasks related to technical, electrical and mechanical installations and maintenance, while white-collar workers work at a desk and perform several types of procedures, write and conclude contracts, perform transactions, etc. While both time and space flexibility are available to white-collar workers, for blue-collar workers smart-work is mainly characterized by being done on a flexible time schedule.

We designed the experiment in agreement with the firm's senior management, who assented to all of our requests and recommendations. We randomized the sample after data had been anonymized by the company. We had complete access to the data used for the analysis and direct access to the surveys administered for the experiment. We also had daily contacts with the managers in charge of the experiment at the company and the management team.

Figure 1 shows the Consolidated Standards of Reporting Trials flow (CON-SORT) that summarizes the flows of the experiment⁵. We adopt the intention-

⁴During 2017 and 2018, smart-working was also one of work-life balance measures that gave companies rights to a tax relief.

⁵Randomized controlled trials can be affected by two major complications: noncompliance and missing outcomes (Gupta, 2011). The more suitable approach is to perform an intention-to-treat (ITT) analysis, which includes all randomized subjects. It ignores noncompliance, protocol deviations, withdrawal, and anything that happens after randomization. The resulting estimate of the treatment effect is generally conservative because of dilution due to noncompliance. Full reporting of any deviations from random allocation and missing responses is essential in the assessment of the ITT approach, as emphasized in

to-treat approach (ITT), which we now explain step by step.

First, we extract our sample of analysis from the population of 4131 workers. We choose to oversample women, workers younger than 46, workers with children under the age of three and workers with relatives who need special care, which gives them the right to reduce the number of hours worked, according to Italian Law 104/92 (Legislation on Support for the Disabled). In agreement with the firm, we decided to oversample these groups because survey evidence suggests that women and individuals with family care duties (caring for children, disabled relatives, etc.) are expected to need and benefit more from smart-working (see Giammatteo (2009)). In fact, the firms was particularly interested in improving working conditions of these categories and their work-life balance, expecting this to reflect also into their productivity.

Using these criteria, we selected 345 workers and asked them about their willingness to join the program. The proportion of those who did not agree was 10%, which in experiments of this type is considered a reasonable number (Jacobsen et al., 2012, Jørgensen et al., 2014). As a consequence, our final sample consists of 310 workers. Among them, 86% are white-collar workers and 14% are blue-collar workers. There are no seasonal workers. Table 1 summarizes the characteristics of our workers, and compares them with those of the total population of workers at the firm. Appendix E provides detailed information on the job description of each worker in our sample, covering both white-collar and blue-collar workers.⁶

Afterwards, we randomly split the above 310 workers into two subgroups, consisting of 200 and 110 workers, respectively: the first group was to engage in smart-working (the treated group) and the second continued to work according

CONSORT (Moher et al., 2001)

⁶Table E.3 in Appendix E illustrates the responses of white- and blue-collar workers to questions in the pre-experiment questionnaire related to dimensions considered crucial for flexible jobs (see Goldin (2014)): time pressure (proxied by the answer to the question "do you comply with the predetermined deadlines of your responsibilities at work?"), contact with others and interpersonal relationships (proxied by the answer to the question "do you feel like having a useful role in your work life?") and freedom to make decisions (proxied by the answer to the question "do you feel capable of making decisions?").

to the preexisting arrangements (the control group).⁷

Based on the experience of other companies that have already implemented smart-working and after an assessment of the company, we choose to define the treatment as one day of smart-working (flexible place and time of work, chosen by the worker) per week, not allowed to be subdivided, and to be planned in agreement with the company on a weekly basis for 9 months starting on October 2016. Given their flexibile schedule during this day, workers' availability is guaranteed within the working day. The day of smart-work does not need to be the same for all workers and can change from one week to another. However, most workers (98 %) choose Friday as a smart-work day. The total number of smart-working days amounts to 4595 days for 200 employees involved and 130 supervisors.

Smart-workers use the same IT equipment, perform the same tasks, and are compensated under the same pay system as are workers belonging to the control group. The only difference between the two groups is the flexible arrangement of time and place for one day per week. Computer equipment was provided for the duration of the treatment to treated group's workers who did not have it. The respective workers knew that they would have the computer only for the duration of the experiment; thus, we do not expect this to affect their perceived status.

The randomization of workers, performed by us in agreement with the firm, is based on gender (male and female), age groups (27-45 and 46-68) and type of job (white-collar and blue-collar). The combinations of these characteristics result in 8 strata. For each stratum, we randomly assign 65% of the individuals to the treated group and the remaining 35% to the control group.

In Table 2, we compare treated and control groups' workers. The table confirms that the two groups are not significantly different in any of the observable characteristics, namely, gender, age, whether the worker benefits from Law 104 or has a relative who benefits from it, whether the worker has children and if

⁷As each treated worker had to be provided with a computer and a maximum of 200 computers were available due to budget constraints, we decided to maximize the number of treated workers. As a consequence, treated and control groups had different sizes.

he/she has children younger than 3 years old.

Within the treatment group, 191 workers received the treatment, while 9 did not receive it: only 5 workers declined to participate after their random allocation into the treatment group, 3 workers died, and 1 worker retired. Within the control group, 2 workers left during the experiment, 1 worker was fired and 1 worker declined to participate. These very small numbers do not pose a concern for the validity of the experiment. Instead, the attrition rate is important, i.e., the failure to follow up: 18 of treated group's workers were not observed after the treatment because they did not reply to the second questionnaire. The respective number increases to 43 for control group's workers. To ensure that the attrition rate is not problematic in our experiment⁸, in Table 3 we present balance tests for observable characteristics of respondents and non-respondents to the post-treatment questionnaire. There are no significant differences, apart from a small difference in gender. In Table 4, we also show balance tests for the intention to treat, i.e. means of the same observable characteristics of workers respondent after the treatment and those who did not, separately for the treated and control groups. Again, we observe no significant differences. In the next section, we will introduce our outcome variables and describe balance tests of pre-treatment outcomes for treated and control workers, including a comparison between respondents and non-respondents to the post-treatment questionnaire to exclude biases from the attrition.

Each worker is matched with his/her supervisor. There are 130 supervisors because some of them supervise more than one worker. Ten supervisors also participate in the experiment, of which 8 in the treated group and 2 in the control group. Among supervisors, 75% are male, and the rest are female. For both male and female supervisors, we have an equal split between the two age groups: under 46 and 46 or above.

During the experiment, workers did not know whether the treatment would be temporary; this prevents possible anticipation effects. Interestingly, after the

 $^{^{8}}$ In a comprehensive review of methodological issues related to presence of attrition rates, Akl et al. (2012) shows that balance tests of respondents and non respondents are appropriate to reduce estimation biases and minimize the lack to follow-up

end of the experiment, given the positive results obtained, the company informed us of its decision to continue and gradually expand the implementation of smart-working. In this subsequent process, the firm gave priority to workers who did not participate in our experiment. Only 60 workers of our sample of 310 workers had the opportunity to apply for participation in the new round of smart-working, of which 80% belonged to our treatment group.

During the nine months, the experiment was constantly monitored through meetings with the company, weekly reporting on the use of smart-working and an internal meeting of the company every 3 months with the target population.

The protocol used has been registered in the American Economic Association's Randomized Control Trial registry. The experiment has been approved by the Ethics Committee at Bocconi University.

4 Data collection

Data are obtained from both the firm and questionnaires administered to workers in both the treatment and control groups and their supervisors before and after the treatment. The firm provides some baseline information for each worker in our sample, as has already been mentioned, namely, gender, age, number and ages of children, whether the worker or a relative needs special care according to Law 104 and the type of work performed by the worker. We also know whether the individual works in a team with other coworkers and with a common supervisor.

4.1 Objective Productivity

Measuring productivity is a difficult task, especially for workers who are not physically present at the workplace. We thus decide to rely on three different measures: objective productivity, self-reported and reported by supervisors. In this section, we describe our measure of objective productivity, while the other two will be introduced in the next section.

The firm provides monthly days of leave (i.e., days off, sick leave, or special leave), the number of hours worked, and a monthly numeric indicator of objective productivity for each worker, built from the results of each worker in his/her own tasks. This is a monthly indicator collected for the entire duration of the experiment from the month prior to testing (September 2016) to the last full trial month (June 2017).⁹ This indicator computed by the firm is unavailable for only 6% of workers in our sample. For the remaining workers, the indicator used is a specific number that represents the respective worker's level of performance. More specifically, for 84.5% of them, the indicator corresponds to an absolute number, while for the other 15.5%, it is measured as a numeric change with respect to the measurement of the previous period. The measurement is homogeneous over time for the same individual but varies across individuals, as it depends on the specific job of each worker. Both the absolute number and the change reflect the exact number of executed tasks (e.g., the number of procedures completed, the number of contracts concluded, the number of transactions performed, etc.), or the evaluation of the employee on a scale reported by the supervisor, or the compliance with deadlines (yes or no).

Since we are interested in the effect of treatment over the entire period (which ensures that the effect does not depend on possible periods of peaks in job duties during the year), we consider cumulative variations of the objective indicator as follows. We consider each monthly variation; the variations can be positive, negative or null. We sum all variations and create a dichotomous variable that has the value of 1 if that sum is positive, thus indicating an overall improvement of productivity over the treatment period, and is zero otherwise. Subsequently, we perform a logistic regression for this dichotomous variable.

For example, consider a worker who has to perform a set of procedures. His/her

⁹According to our knowledge of the operation of the firm and information received by the managers, September is an ordinary month in terms of productivity of workers. As the new working arrangements may take some time to fully stabilize, we also rerun the analysis, considering October 2016 as the month preceding testing. This adjustment does not change results.

productivity is measured on the basis of how many procedures he/she is able to complete during a month. In the reference month of September, he/she completed 68 procedures, while the respective number was 14 in October, 79 in November, 105 in December, etc. We calculate for each month from October to June the difference as a percentage with respect to the value as of September. Next, we sum all of these percentages. A positive value of this final sum indicates an improvement of productivity over the examined period. If there are more tasks assigned to the same worker, we consider the overall sum for all tasks. We proceed similarly if a task is measured by the number of concluded contracts or that of performed transactions or another quantitative metric. As another example, consider a worker who receives an evaluation on the scale from 1 to 10 every month. If the evaluation in the reference month of September is equal to 6 and in the following month is equal to 8, the change in the latter month is positive and equal to 2. We sum the changes between the evaluation of each month and the evaluation of September and create a dichotomous variable that has the value of 1 if that final sum is positive and is 0 otherwise.

We present several figures that visually show the evidence we will analyze. Figure 2 shows for each month of the treatment period the percentage of treated and control groups' workers whose productivity increased relative to that in the previous month. In each month, the percentage for treated group's workers is higher than that for the control group's workers. This suggests that the difference between treatment and control is not only an impact of the introduction of flexible work, probably related to the change, but continues over time.

To understand whether the change in productivity varies over time during the experiment, in Figure 3 we show the objective productivity of treated and control groups' workers after 3, 6 and 9 months. The difference is not statistically significant after 3 months but appears after 6 months.

Finally, Figure 4 compares the average numbers of days of leave requested monthly by treated and control groups' workers over the period of the experiment. Interestingly, starting from the same point, after one month the total amount of days of leave of the treated group is consistently lower than that of the control group.

4.2 Questionnaires

We designed questionnaires to collect data on productivity, well-being and the work-life balance. Workers and supervisors in both the treated and control groups were asked to complete one questionnaire before the experiment (the pre-treatment questionnaire) and another one afterwards (the post-treatment questionnaire). The questionnaires include questions related to productivity, well-being and the work-life balance. In the post-treatment questionnaire, the treated group also answered questions related to their evaluation of the policy. Questionnaires are shown in Appendix F.

Productivity assessed by the questionnaires includes self-reported productivity and that reported by supervisors. Self-reported productivity includes 5 outcomes: a measure of output, i.e., the capacity to attain the assigned goals, efficiency at work, i.e., the capacity to attain the assigned goals within an appropriate time, proactivity at work, i.e., the capacity to take initiative appreciated by others; availability to answer emails or work outside working hours and capacity to comply with predetermined deadlines. The respondents are asked to evaluate each outcome on a scale from 1 to 5, where 1 corresponds to "Very Low", and 5 corresponds to "Very High". Productivity reported by the supervisor includes the same 5 outcomes. They are consistent with measures used by existing case studies, reports, and toolkits focused on flexible work and productivity (see Golden (2012), Pruchno et al. (2000), Kossek and Michel (2011), etc.).

The well-being assessment includes standard questions drawn from the British Household Panel Survey (Taylor et al. (1993)). These questions are widely used in the literature on economics of happiness (see reviews in Van Praag et al. (2003) and Luhmann et al. (2012)). Respondents are asked to indicate the extent of their satisfaction on a scale from 1 to 7, where 1 corresponds to "highly dissatisfied" and 7 corresponds to "highly satisfied" with the following 7 dimensions: income, health status, home, job, social life, free time, and life overall. Respondents also have to report on a scale from 1 (much less than usual) to 5 (much more than usual) their ability to deal with the following 7 aspects of their lives: staying focused (referred to as "FocusedOn" in the tables), losing sleep due to any concerns, feeling that they play a useful role in their work life, being able to make decisions, appreciating the daily activities in a regular day, feeling stressed, and feeling unable to overcome difficulties.

The work-life balance assessment asks about satisfaction with working hours, how working hours adapt to private life and the feeling of being able to balance work with personal and family life. Workers are also asked to quantify the time devoted to household activities per day (cleaning and housekeeping) in two-hour ranges from "Less than 2 hours" up to "More than 6 hours" and to quantify the time dedicated to taking care of others (children, elderly, or other family members).

We provide first insights on the well-being and work-life balance indicators through a visual analysis. Figure 5 and Figure 6 display kernel density functions for the treated and control groups before and after the treatment with reference to two specific outcomes: satisfaction with social life and ability to deal as usual with stress. Interestingly, while the pre-treatment kernel densities of treated and control groups overlap almost completely, they diverge after treatment, which thus suggests the emergence of an effect of the intervention.

The questionnaires also ask for the home-to-work distance in kilometers.

Finally, they contain questions about the commitment of workers to the company. The questions are: "How attached do you feel to the company?", "Do you believe your work is sufficiently recognized?" and "Do you have a sense of moral responsibility towards the company?"

Table 6(panel a) shows the results of the t-test for the difference between the means of treated and control groups for each outcome variable measured before the treatment (i.e., using data from the pre-treatment questionnaire). It confirms that the starting point is the same for treated and control groups, i.e., the two groups are balanced before the treatment, which confirms the validity of our randomization. The same result is obtained when we consider only respondents (in both treated and control group) to the post-treatment questionnaire (panel b). To further insure that the attrition rate is not problematic for our results, we also show balance tests of pre-treatment outcomes for respondents and non respondents (to the post-treatment questionnaire) within the treated group (panel c) and the control group (panel d).

Table 7 summarizes the outcome variables from the post-treatment questionnaires, related to the different dimensions of productivity, well-being, work-life balance and commitment to the company.

Additionally, we check that the difference between the number of hours worked by treated and control groups' workers before and after the experiment is not statistically significant, confirming that any possible effect of treatment on the outcome variables does not depend on a change in the number of hours worked (Table 5).

5 Empirical strategy

We estimate the effect of treatment (i.e., of smart-working) on our indicators of productivity, well-being and work-life balance.

One of the measures of productivity is the improvement of objective productivity, a dichotomous outcome that has the value of 1 if the sum of the monthly productivity changes is positive, and is zero otherwise. We run a logistic regression to estimate this outcome and report the odds ratio of improvement for treated group's workers compared to those of the control group.

We use an ordinary least squares (OLS) model to estimate the effect of treatment on the other indicators of productivity (number of days of leave, selfreported productivity and productivity reported by the supervisor) and on the indicators of well-being and work-life balance. For each of these indicators, we estimate the following equation at the individual level:

$$y_{i,POST} = \alpha + \beta TREATMENT_i + \delta X_i + \gamma y_{i,PRE} + \varepsilon_i \tag{1}$$

where $y_{i,POST}$ is the specific measured outcome for individual *i* post-treatment according to the dimensions of productivity, well-being and work-life balance, as summarized in Table 7, $TREATMENT_i$ is a dummy variable that has the value of 1 if individual i has been assigned to the treatment group and is 0 if he/she belongs to the control group, $y_{i,PRE}$ is the specific outcome for individual i measured before treatment according to the three mentioned dimensions (productivity, well-being and work-life balance), X_i are individual control variables, and finally, ε_i is an error term. Given the randomization, there should be no need to add control variables to measure the average treatment effects. However, control variables are included to improve the accuracy of estimates. Control variables are: the age of respondent (AGE) and its square (AGESQUARED), gender, captured by a dummy variable MALE that has the value of 1 if the respondent is male and 0 if the respondent is female, two dummy variables (LAW104WORKER) and LAW104RELATIVES) that capture the use of Law 104 for the worker or for a relative, respectively, two dummy variables related to children, namely, CHILD that has the value of 1 if the worker has at least one child and is 0 otherwise, and YOUNGCHILD that has the value of 1 if at least one of the children of the worker is aged 3 or below and is 0 otherwise, a dummy variable TEAM that is equal to one if the worker works in a team and is 0 otherwise, and the distance between the worker's residence and the workplace in kilometers (KM).

Since, as has already been mentioned, we oversample some characteristics of the population (female, people under the age of 46, subjects with children under the age of three and workers with a relative protected by Law 104), using weighting is recommended to make the analysis of the sample representative for the target population. We thus create a set of weights that is the inverse of the probability of inclusion of each stratum of the oversampled categories. In what follows, we report all analysis results using these weights.

We also use a difference-in-differences model to estimate the effect of treatment on the treated group, as opposed to the control group. The model and the results of these estimates are reported in Appendix D. They are consistent with what is presented in the following sections of the main text.

6 **Results**

We present results according to the three dimensions we investigate: productivity, well-being and work-life balance.

6.1 Productivity

The first dimension we consider is productivity. As explained, we consider three measures of productivity: objective, self-reported by the worker and reported by the supervisor. In each table and for each outcome, the first column reports the average treatment effect, and the second column includes control variables. The coefficients of the control variables are shown in Appendix A.

Starting from the indicator of objective productivity, Table 8 panel a, columns 1 and 2 show a significant increase of objective productivity for the treated group after treatment. The odds ratio for treated workers is twice the odds ratio for the control group. Second, we consider days of leave. Results of OLS estimates of Equation 1 are reported in Table 8 in columns 3 and 4. The treatment reduces the number of days of leave (the reduction is by 5.6 days over the entire period of the experiment). This likely occurs because smartworkers can better organize their time than can non-smart-workers and are less in need of asking for leave due to sickness or other reasons (e.g., if they need to visit a doctor or have to pick up children from daycare centers or assist elderly parents, etc.).¹⁰

Overall, panel a of Table 8 suggests an increase of productivity associated with smart-working. What drives this increase in productivity? It is difficult

¹⁰Note that for objective measures we work with a sample size that differs from that for the other outcomes. This is due to the different source of data, which is obtained directly from the firm rather than from questionnaires. The percentage of missing data points is higher because the analyzed sample has different missing values from those in the information supplied by the firm.

to test exactly what happens to smart-workers that has a positive effect on their objective productivity. It might be that smart-workers spend less time at lunch or on coffee breaks or bathroom breaks, as it has also been observed for telecommuters (see Bloom et al. (2014)). Other possible reasons that could explain the increase of productivity of smart-workers relate to the time saved in commuting or in taking children to and picking children up from school, which could allow workers to start working earlier, or to be more focused on their jobs during the smart-working day. However, we cannot appropriately test for this.

Another interesting possibility is that, due to smart-working, workers become more attached to the company, and thus are also more productive. According to the "Hawthorne effect", treated workers have a positive feeling toward the firm, which allows them to use smart-working and reciprocate by working harder (see Falk and Kosfeld (2006)). In other words, they work more efficiently because they feel an obligation to the company. To test the existence of this effect, we rely on the information on workers' commitment to the company. Table 16 reports the results of estimating equation 1 using as the dependent variable the answers to the following questionnaire's questions: "How attached do you feel to the company?", "Do you believe your work is sufficiently recognized?" and "Do you have a sense of moral responsibility towards the company?" Results show that treatment is associated with a greater sense of moral responsibility. The other two dimensions of commitment are instead not statistically significant. These results indicate some evidence of the existence of the "Hawthorne effect".

Additional insights into the mechanisms at work will be drawn from the analysis of heterogeneous effects across workers with different characteristics, which we explore in sections 6.4 and 7.

Next, we consider self-reported productivity that includes the five outcomes described in section 4.2: output ("production"), efficiency, proactivity, availability to respond to emails, and compliance with deadlines. Panel b of Table 8 shows results of estimating equation 1 obtained by using these five measures as the dependent variable. The average treatment effect is positive and significant for all measures except availability to respond to emails, and remains significant (apart from productivity) if control variables are included.¹¹

Finally, we estimate equation 1 using as the dependent variable the same five outcomes while instead using the answers of the supervisor of each worker. The respective results are shown in panel c of Table 8, which confirms that smart-workers increase their compliance with deadlines compared to results for the control group; this finding also holds if workers are assessed by supervisors. We note that the number of observations in this analysis is smaller because supervisors did not always respond.

6.2 Well-being

The second dimension we consider is well-being. As anticipated, we measure well-being in two ways. First, workers are asked to self-assess on a scale from 1 to 7 their personal satisfaction with respect to seven variables: income, health status, home, work, social life, free time, and life in general. Second, they are asked whether they are able to deal as usual on a scale from 1 to 5 (where 1 corresponds to "much less than usual" and 5 to "much more than usual") with seven aspects of their life: staying focused, losing of sleep due to any concerns, feeling that they play a useful role in their work life, being able to make decisions, appreciating the daily activities in a regular day, feeling stressed, and feeling unable to overcome difficulties.

Panel a of Table 9 shows the results of estimating equation 1 using as the dependent variable seven different measures of satisfaction, while panel b shows results for seven "satisfaction as usual" measures of well-being. Smart-working increases the individual satisfaction with social life, free time and life in general. When we include control variables, positive and significant effects are also observed for satisfaction with income, health and home.

Moreover, smart-workers are more capable than usual of dealing with all aspects of their lives (apart from playing a useful role): in panel b of Table 9,

¹¹Note that, consistently with most of the literature on this topic, age has a nonlinear relationship with (self-declared) productivity: the latter increases up to a certain age and subsequently starts to decline.

the coefficients of treatment are positive and significant.¹²

6.3 Work-life balance

The third dimension we consider is the work-life balance, measured by four variables that correspond to satisfaction with four aspects: working hours, balance between working life and personal/family life, the amount of house-hold activities (cleaning and housekeeping) per day and the amount of time dedicated to taking care of others (children, elderly, or other family members).

Panel a of Table 9 reports estimates of equation 1, where for each column the dependent variable is one of the above 4 measures of work-life balance. The table shows that treatment is associated with more time being dedicated to household and care activities.

6.4 Heterogeneous effects by gender

One of the most interesting yet expected consequences of the introduction of flexible work is the reduction of gender gaps (Goldin, 2014). Smart-working is expected to reduce gender gaps for at least two reasons. First, although smart-working does not target women, it may be particularly promising for women's employment because it promotes work-family balance, which is a major concern for employed women, who typically bear the double burden of work and family/care responsibilities. This is particularly true in Italy, where, according to the most recent data of the Italian National Institute of Statistics (Istat, 2018), women spend 3 hours per day more than men in domestic and unpaid care work, and more than 4 hours if we consider couples with children. The country exhibits the highest asymmetry in time use within couples across all European countries. Second, by promoting work-life balance and a more

¹²In Appendix A, we report the coefficients of covariates. Note that being a woman is significant when we measure satisfaction with income. As intuitively expected, age and the use of Law 104 are instead negatively related to satisfaction with health. Interestingly, workers with children are more satisfied with their income and health, but those with a young child are less satisfied with income and health. Workers who live farther from their workplace are more satisfied with their home and work.

efficient allocation of time, smart-working may increase the participation of men in housework and childcare.

In this section, we examine heterogeneous effects for men and women in our sample. Our balance tests for the covariates by gender, presented in Table 10, confirm that the subsamples of men and women are randomly divided, and thus our analysis by gender is expected to be informative.

Table 11 shows that the improvement of objective productivity and a reduction of the number of days of leave is driven by women. However, if productivity reported by supervisors is considered, the increase in availability and improved compliance with deadlines (the two dimensions that were overall significant) is more likely attributed to men. Table 12 shows that male smart-workers show a higher satisfaction with home, social life and free time, while female workers exhibit higher satisfaction with work and life in general. Considering the self-assessed comparison with the usual conditions, women feel more focused, lose less sleep and feel more useful than before the treatment (table 12, panel b). Yet men show a significant improvement in being able to "focus on" and appreciate daily activities, experiencing less stress and feeling more able to overcome difficulties (table 12, panel b). Interestingly, considering the indicators of work-life balance (table 12, panel c), we observe that men claim to spend significantly more time in household work and care activities after the introduction of smart-working. In other words, as anticipated, and against the stereotype that men may use job flexibility for performance purposes and women for work-family balance, we find causal evidence that smart-working increases participation of men in household and care activities, which is a fundamental step towards more gender equality. Women are more satisfied with working hours and the impact of work on women's private life. Smart-working contributes to the reduction of gender gaps through both a better work-life balance for women and greater participation of men in housework and care activities.

This result suggests that smart-working should be particularly appealing to workers with children. As the treated and control groups are balanced with respect to the number of children (see Table C.1 in Appendix C) in addition to other factors, we investigate heterogeneity between workers with and without children. We observe no heterogeneous effects in productivity. The results seem to suggest that, as expected, while smart-working is beneficial for the well-being of all individuals, it is particularly appreciated by workers with children who have strong work-life balance needs (see Tables C.3 and C.4 in Appendix C).¹³

7 Heterogeneous and spillover effects by team

Other interesting yet expected consequences of the introduction of flexible work are the different effects of smart-working for workers in teams and workers not in teams and the possible spillover effects from treated group's workers to control group's coworkers in the same team. The literature, though limited to low-skill jobs, suggests that workers who observe their peers increase their efforts and productivity (Mas and Moretti, 2009). Thus, smart-working, by introducing one day of remote and flexible working, may produce a negative effect on productivity of control group's workers of the same team. This negative spillover effect characterized also telecommuting (Bloom et al. (2014)).

In this section, we first consider working in a team as a dimension of heterogeneity. We know whether each worker – both in the treated group and in the control group – works in a team and are able to identify the coworkers in the same team. Approximately 70% of workers work in teams. The characteristics of workers in the two subgroups – of those working in teams and of those not working in a team – are balanced across the treated and control groups (see Table 13).

Working in a team does not seem to be significantly related to productivity measures (Appendix A). However, we are interested in possible differential effects of smart-working for workers in teams and workers not in a team. In

¹³Our randomized experiment allows us to also analyze heterogeneous effects by age and type of job, for which, however, we do not have clear expectations. This analysis is available upon request. Unfortunately, we are unable to analyze heterogeneous effects by work-to-home distance because there is no division in balanced subgroups according to this dimension.

Table 14, we add the team variable and the interaction between team and treatment to the baseline scenario. Smart-working increases the objective productivity of workers but does not do so differently for workers in teams and workers not in a team. Self-reported productivity and productivity reported by supervisors are unaffected by working in a team, showing that there are no significant spillover effects of productivity on team workers. A different result is obtained for the number of days of leave, which significantly increases for smart-workers in teams. Treated group's workers not in a team show a decrease by more than 18 days in the number of days of leave, which is reduced to 1.154 days (17.825-18.979=-1.154) for treated workers in teams. The latter seem to be able to enjoy more days of leave than do those not in a team, without altering their objective productivity.

Second, we consider the possible spillover effects from treated group's workers to control group's workers of the same team. We consider workers in the control group and compare the performance of those who had at least one member of their team included in the treated group and that of the others. In panel a of table 17, we do not observe a significant difference of self-reported productivity for these two groups. No significant difference is detected either if we compare workers in the control group in a team with more than 40% of treated group's members and the other workers (panel b). Similarly, we consider the subsample of individuals in the control group and compare their outcomes before and after the treatment of their colleagues (Table 18). Again, we consider control workers in a team with at least one treated group's workers (panel a) and those in a team with more than 40% of treated group's workers (panel b). Both panels show that there are no significant negative spillover effects.¹⁴

8 Additional analysis and Robustness tests

We perform several robustness checks to address potential concerns as to the interpretation of our results.

¹⁴Overall, the absence of spillover effects is a validation of our experiment because it confirms that treatment and control groups are clearly separated.

A typical concern in research using questionnaires relates to the reliability of answers when questionnaires are quite long, as in our case. Scholars of survey methods have underscored that when individuals answer a long questionnaire, at a certain point they often start answering automatically (see Pintrich et al. (1991)). For this reason, we introduced several questions about a similar topic but using a reverse scale. The obtained results reassure against this concern: respondents answered carefully, as questions with a positive meaning but asked in a negative way were correctly answered negatively.

Since our dataset is very rich and allows us to include a large number of control variables that could potentially influence the dependent variables, we use the method of stepwise selection of covariates to select the best model to estimate. Results, available upon request, are very similar to what we presented in our main tables.

As a further analysis, we perform the same regressions without weights. Similar results are obtained. This demonstrates stability of our results and confirms that the extracted sample is representative of the firm's workers even though we have over- or under-sampled some characteristics. Results of the unweighted analysis are available upon request.

Another typical concern with a randomized experiment of this type is that it is difficult to identify whether the significant effects observed for the treatment group depend on the improvement of the treated group or a worsening performance of the control workers who are dissatisfied with having being excluded from the treatment, though randomly. Since we still observe a performance gap between treatment and control groups – in productivity, well-being and work-life balance dimensions – our estimated treatment effect is likely to be downward-biased.

As a further test, in table 19 we consider various features of the control group and compare for each feature its average performance before and after treatment even if control group's workers had not been subject to it. The t-tests show that there are no statistically significant differences in the outcomes of the control group before and after treatment, apart from three outcomes (compliance with deadlines, household activities and care activities). Yet, for these three outcomes the direction of the change is positive, i.e., workers in the control group increase rather than decrease their performance after the policy's implementation. Thus, the increased productivity of the treated group's workers cannot be confounded by a worsening performance of the control group.

Finally, to confirm our results, we produce bounds for the estimates of several outcome variables as proposed by Lee (2009) (Appendix B). We consider the outcomes on which smart-working has a strong significant impact: satisfaction with social life, free time and life in general, feeling "as usual" with respect to staying focused and appreciating daily activities, losing less sleep and feeling less stressed, and experiencing better work-life balance of household and care activity. The upper and lower bounds of these estimates have the same sign, and 95% confidence intervals for the bounds only barely include zero.

9 Discussion and Conclusion

We have established a causal link between the introduction and use of smartworking and several economic outcomes that capture workers' productivity, well-being and work-life balance. The three sets of outcomes together suggest interesting insights.

First, smart-workers claim to be more satisfied with their free time and social life. Does this mean that they reduce the extra paid work hours and thus their earnings? Although we do not have data on earnings to directly test this effect, we note that smart-workers also claim to be more satisfied with their incomes, suggesting that there is no negative effect of smart-work on earnings.

Second, the observed increase in productivity means that, for the same pay, smart-workers put more effort into their jobs than do non smart-workers. Smart-workers are more focused and more active. This may be the result of different and more effective organization of their time, including a reduction of commuting time and a better use of time within the household. This effect is captured by the work-life balance indicators that show that smartworkers spend more time in household and care activities. We also note that the corresponding increase in well-being indicators suggests that the increase in productivity goes beyond different and better time management. The fact that job satisfaction increases even if workers apply more effort means that smart-workers have a positive perception of the new form of work organization: they are ready to exchange more effort for more flexibility to maintain or even increase their job satisfaction.

Our study will have substantial policy implications. Smart-working is a recent approach that is rapidly spreading and is now regulated in several countries. Removing the constraints of space and time of work looks a promising way for a more efficient organization of working. Moreover, our results are stronger for women. Consistently with the view of Goldin (Goldin, 2014), our results suggest that the flexibility introduced by smart-working may contribute to reducing gender gaps at work.

Smart-working also appears to be a promising way to promote work-life balance, which is becoming a significant issue in modern societies. Interestingly, this result was not obvious, as previous analyses have warned about the risk of over-working related to flexible work arrangements, with all the possible negative consequences (involving stress, well-being, health, etc.). Ex-post evaluations such as those performed by previous studies are, however, unconvincing, as they may hardly infer causality. Random assignment as in the approach we used in our study is a way to guarantee the direction of causality and unbiased estimates. In other words, having based our analysis on a randomized experiment, we can ensure that we have established an internally valid identification of the effects of smart-working.

The spread of coronavirus is accelerating the use of smart-working as the only way for firms to respond to this disruptive event. This will also be an opportunity to reimagine work organization after the crisis. In this case, our results will have useful applications.

It is difficult to assess the extent to which our results can be generalized to other contexts. This is a common concern for all randomized field trials in economic and policy research. However, our experiment was performed in a developed country, for a large set of jobs, and in a context quite representative of the current conditions in many other countries. While this is the first study of this type, future studies will assess whether the same results apply also to other contexts, where smart-working is indeed discussed and implemented, but no causal evidence of its economic effects exixts.

References

- Akl, E. A., Briel, M., You, J. J., Sun, X., Johnston, B. C., Busse, J. W., Mulla, S., Lamontagne, F., Bassler, D., Vera, C., et al. (2012). Potential impact on estimated treatment effects of information lost to follow-up in randomised controlled trials (lost-it): systematic review. *Bmj*, 344:e2809.
- Bandiera, O., Barankay, I., and Rasul, I. (2005). Social preferences and the response to incentives: Evidence from personnel data. *The Quarterly Journal* of Economics, 120(3):917–962.
- Bertrand, M. (2018). Coase lecture–the glass ceiling. *Economica*, 85(338):205–231.
- Bloom, D. E. and Sousa-Poza, A. (2013). Ageing and productivity: Introduction. *IZA Discussion Paper*, (No. 7205).
- Bloom, N., Liang, J., Roberts, J., and Ying, Z. J. (2014). Does working from home work? evidence from a chinese experiment. *The Quarterly Journal of Economics*, 130(1):165–218.
- Chung, H. (2011). Work-family conflict across 28 european countries: A multilevel approach. In *Work-Life Balance in Europe*, pages 42–68. Springer.
- Chung, H. and Meuleman, B. (2017). European parents attitudes towards public childcare provision: The role of current provisions, interests and ideologies. *European Societies*, 19(1):49–68.
- Dean, A. and Auerbach, A. (2018). 96 percent of us professionals say they need flexibility, but only 47 percent have it. *Harvard Business Review*.
- Dutcher, E. G. (2012). The effects of telecommuting on productivity: An

experimental examination. the role of dull and creative tasks. Journal of Economic Behavior & Organization, 84(1):355-363.

- Eurofound (2017). Working anytime, anywhere: The effects on the world of work.
- Falk, A. and Ichino, A. (2006). Clean evidence on peer effects. Journal of labor economics, 24(1):39–57.
- Falk, A. and Kosfeld, M. (2006). The hidden costs of control. American Economic Review, 96(5):1611–1630.
- Gallup, I. (2017). State of the american workplace.
- Giammatteo, M. (2009). L'indagine campionaria Isfol-PLUS: contenuti metodologici e implementazione. *Studi Isfol*, 3.
- Gibbons, R. and Henderson, R. (2013). Relational contracts, managerial practices, and organiza-tional capabilities. *Handbook of Organizational Economics*.
- Golden, L. (2012). The effects of working time on productivity and firm performance, research synthesis paper. International Labor Organization (ILO) Conditions of Work and Employment Series, (33).
- Golden, L. and Wiens-Tuers, B. (2006). To your happiness? extra hours of labor supply and worker well-being. *The Journal of Socio-Economics*, 35(2):382–397.
- Goldin, C. (2014). A grand gender convergence: Its last chapter. American Economic Review, 104(4):1091–1119.
- Gupta, S. K. (2011). Intention-to-treat concept: a review. Perspectives in clinical research, 2(3):109.
- Halbesleben, J. R. and Buckley, M. R. (2004). Burnout in organizational life. Journal of management, 30(6):859–879.
- Hegewisch, A. et al. (2009). *Flexible working policies: a comparative review*. Equality and Human Rights Commission Manchester.

- Jacobsen, P. B., Wells, K. J., Meade, C. D., Quinn, G. P., Lee, J.-H., Fulp, W. J., Gray, J. E., Baz, R. C., Springett, G. M., Levine, R. M., et al. (2012). Effects of a brief multimedia psychoeducational intervention on the attitudes and interest of patients with cancer regarding clinical trial participation: a multicenter randomized controlled trial. *Journal of Clinical Oncology*, 30(20):2516.
- Jørgensen, R., Munk-Jørgensen, P., Lysaker, P. H., Buck, K. D., Hansson, L., and Zoffmann, V. (2014). Overcoming recruitment barriers revealed high readiness to participate and low dropout rate among people with schizophrenia in a randomized controlled trial testing the effect of a guided self-determination intervention. *BMC psychiatry*, 14(1):28.
- Kelly, E. L. and Moen, P. (2007). Rethinking the clockwork of work: Why schedule control may pay off at work and at home. Advances in developing human resources, 9(4):487–506.
- Kelly, E. L., Moen, P., Oakes, J. M., Fan, W., Okechukwu, C., Davis, K. D., Hammer, L. B., Kossek, E. E., King, R. B., Hanson, G. C., et al. (2014). Changing work and work-family conflict: Evidence from the work, family, and health network. *American Sociological Review*, 79(3):485–516.
- Kelly, E. L., Moen, P., and Tranby, E. (2011). Changing workplaces to reduce work-family conflict: Schedule control in a white-collar organization. *American Sociological Review*, 76(2):265–290.
- Koopman, C., Pelletier, K. R., Murray, J. F., Sharda, C. E., Berger, M. L., Turpin, R. S., Hackleman, P., Gibson, P., Holmes, D. M., and Bendel, T. (2002). Stanford presenteeism scale: health status and employee productivity. *Journal of occupational and environmental medicine*, 44(1):14–20.
- Kossek, E. E. and Michel, J. S. (2011). Flexible work schedules.
- Lee, D. S. (2009). Training, wages, and sample selection: Estimating sharp bounds on treatment effects. *The Review of Economic Studies*, 76(3):1071– 1102.

- Leibenstein, H. (1966). Allocative efficiency vs." x-efficiency". The American Economic Review, 56(3):392–415.
- Leslie, L. M., Manchester, C. F., Park, T.-Y., and Mehng, S. A. (2012). Flexible work practices: A source of career premiums or penalties? Academy of Management Journal, 55(6):1407–1428.
- Luhmann, M., Hofmann, W., Eid, M., and Lucas, R. E. (2012). Subjective well-being and adaptation to life events: a meta-analysis. *Journal of per*sonality and social psychology, 102(3):592.
- Mas, A. and Moretti, E. (2009). Peers at work. *American Economic Review*, 99(1):112–45.
- Mas, A. and Pallais, A. (2017). Valuing alternative work arrangements. American Economic Review, 107(12):3722–59.
- Moen, P., Kelly, E. L., Fan, W., Lee, S.-R., Almeida, D., Kossek, E. E., and Buxton, O. M. (2016). Does a flexibility/support organizational initiative improve high-tech employees' well-being? evidence from the work, family, and health network. *American Sociological Review*, 81(1):134–164.
- Moen, P., Kelly, E. L., and Lam, J. (2013). Healthy work revisited: Do changes in time strain predict well-being? *Journal of occupational health psychology*, 18(2):157.
- Moher, D., Schulz, K. F., and Altman, D. G. (2001). The consort statement: revised recommendations for improving the quality of reports of parallel group randomized trials. *BMC medical research methodology*, 1(1):2.
- OECD (2016). Be flexible: background brief on how workplace flexibility can help european employees to balance work and family.
- Pintrich, P. R. et al. (1991). A manual for the use of the motivated strategies for learning questionnaire (mslq).
- Pruchno, R., Litchfield, L. C., and Fried, M. (2000). Measuring the impact of workplace flexibility. Boston College Center for Work & Family, Carroll School of Management.

- Riedmann, A., Bielenski, H., Szczurowska, T., and Wagner, A. (2006). Working time and work–life balance in european companies, establishment survey on working time 2004–2005.
- Schieman, S., Glavin, P., and Milkie, M. A. (2009). When work interferes with life: Work-nonwork interference and the influence of work-related demands and resources. *American Sociological Review*, 74(6):966–988.
- Syverson, C. (2011). What determines productivity? Journal of Economic literature, 49(2):326–65.
- Taylor, M. F., Brice, J., Buck, N., and Prentice-Lane, E. (1993). British household panel survey user manual. University of Essex.
- Van Praag, B. M., Frijters, P., and Ferrer-i Carbonell, A. (2003). The anatomy of subjective well-being. *Journal of Economic Behavior & Organization*, 51(1):29–49.
- Walker, F. A. (1887). The source of business profits. *The Quarterly Journal* of *Economics*, 1(3):265–288.